**Author's response to reviews**

**Title:** A rare cause of small bowel obstruction in a jejunal diverticulum: a case report.

**Authors:**

  mohammad tayeb (drmtayeb@yahoo.com)
  faiz khan (faizkhan.surg@yahoo.com)
  fozia rauf (farauf@yahoo.com)
  mumtaz khan (mmutazkhan@gmail.com)

**Version:** 5  **Date:** 29 May 2011

**Author's response to reviews:**

Cover letter
To
The JMCR Editorial Team
Subject: Making the suggested changes by editorial team in our Article Case report: A rare cause of small bowel obstruction in a jejunal diverticulum: a case report”.

Authors:
mohammad tayeb (drmtayeb@yahoo.com)
faiz khan (faizkhan.surg@yahoo.com)
fozia rauf (farauf@yahoo.com)
mumtaz khan (mmutazkhan@gmail.com)
MS: 1624372875499765

The editorial team suggested to change the title and so we did it according to your suggestion.
The new title is now “A rare cause of small bowel obstruction in a jejunal diverticulum: a case report”.

I am grateful to the editorial team for their quick response, valuable suggestions and help regarding publication of this article.